Author's response to reviews

Title: Non-neuronal and neuronal BACE1 elevation in association with angiopathic and leptomeningeal beta-amyloid deposition in the human brain

Authors:

Zhi-Qin Xue (kxglsjs@126.com)
Zheng-Wen He (hezhw2001@163.com)
Jian-Jun Yu (dryujianjun@sina.com)
Yan Cai (caojicyan@csu.edu.cn)
Wen-Ying Qiu (qiuwy73@126.com)
Aihua Pan (panaihua@csu.edu.cn)
Wei-Ping Gai (weiping.gai@flinders.edu.au)
Huaibin Cai (caih@mail.nih.gov)
Xue-Gang Luo (luoxuegang@csu.edu.cn)
Chao Ma (mcpumc@163.com)
Xiao-Xin Yan (yanxiaoxin@csu.edu.cn)

Version: 3 Date: 5 March 2015

Author's response to reviews:

Dear editor:

Thanks for the instruction. We have included the neuropathological staging information in the patient table (page 35), based on Braak's scoring. This is also briefly noted in the method section (page 5), with two references cited (page 27). Reference order in the text is reset.

Many thanks,

Xiaoxin